Notes of the RICH1 Phone meeting 13/10/2006
People Present:
CERN:  Carmelo, Clara(part), Fabio, 
IC:         Bill, Dave, Gabriel, Richard P.
Oxford:  Neville, Rohan
RAL:      Mike, Peter
Bristol:   Nick, Richard
Apologies for absence:
Derek, Geoff, Kenneth, Trevor
Discussed Agenda Topics:
1. Gas Enclosure
Next week the 0-ring will be replaced. The survey of the quartz window can be done easily without removing anything and be easily organized.
2. Exit Window

The EW was sent to CERN but was difficult to trace it. It was mistakenly sent to another building. Derek arranged to transport it to building 156 on Monday. The installation kit is at IP8. Monday the boss will be glued to the EW and will be left two days to cure. Next Wednesday at 2pm there will be the safety inspection of the EW installation kit. The EW will be installed on Thursday and Friday. There is enough manpower available. 
Diaphragm: New pigment with a 5% mix which passes the light transmission tests was used to manufacture eight diaphragms. The diaphragms were heat tested (250 0C 10 hours stretched/unstretched). All did well and passed the leak test.  It was decided to use Sylgard type diaphragm. The diaphragms will be assembled manually from outside the RICH. The grease to be used is Apiezon-L from Dow Corning.
3. Beampipe
No news. The beampipe is in an aluminium protective case. Bill will meet with Delio at IP8 on Monday to discuss procedure.

4. Quartz Windows

The HEREUS drawings are on EDMS (782860). The quartz panes are at CERN. They are of good quality. The lengths are slightly out of tolerance and the thicknesses of the panes vary but within tolerance. The QW panes have been accepted. The panes fit it in the coating plant and in the reflectometer. The coating is planned for Dec06. In the meantime the four boxes for storing the QW panes and the coating plate need to be manufactured. The uncoated rim to allow gluing is 10mm. The anti-reflective coating is expected to increase the transmission by 3-5%, as measured for the RICH2 panes. 
5. Spherical Mirrors
Irradiation: Spherical Mirror Sample 2 and Flat sample B have been irradiated to a total of 10 kGy at Ionisos (Lyon) at the end of August 06. The D0,R and reflectivity of the mirror and surface roughness was re-measured and found to be unchanged (Sample 2: D0=0.56mm, R=1898mm).  The mechanical tests and microscope photos of the flat sample will be completed next week. The radiation exposure is completed (10kGy ~10 years in LHCb).
C4F10 exposure: The fluorocarbon gas tank was opened on the 10th Oct06. The samples (Spherical Mirror Sample 1, Flat sample A and the demo mirror) were re-measured and found to be unchanged (D0, R, reflectivity, surface roughness). The mechanical tests were not done because the apparatus suddenly broke. The samples are back in the tank (12th Oct06).
CMA sample 1:  D0= 0.75mm, R=1889mm ;   4 months and a week C4F10 exposure 
Demo mirror:      D0=1.04mm,  R=2205mm;    3 months and a week C4F10 exposure 
It has been decided to test the samples exposed to C4F10 every 3 months.
Three uncoated CMA carbon fiber samples have been sent to CERN. A glass witness sample and a CMA sample have been coated with Al+MgF2 at CERN. The reflectivity curve of the coated CMA sample was very good and very similar to that of the glass sample. The coated CMA sample will be uncoated and re-coated to see if the uncoating procedure has any effect on the CMA substrate.
At last CMA phone meeting (4/10/06), CMA reported that the mandrel polishing is nearly completed and all the material is in house and that they are starting to produce the tubular structures.  CMA is still running calibration tests with the small evaporation chamber and in parallel upgrading the big chamber. Some calibration samples will be sent to CERN for testing and their solid model drawings will be made available. The next CMA meeting is Wednesday 18/10/06. Also a phone meeting CMA-Andre is planned for in the coming week (action Fabio).
Timescales and costs for the backup solution of coating CMA mirrors at CERN needs to be investigated (action Fabio).
6. Flat Mirrors
The drawings for the remaining pieces of the second plates are being prepared and once completed the plates will go onto the CNC machine.

A draft drawing for the alignment rig (flat and spherical mirrors) has been prepared by Richard and needs to be discussed (action Fabio, Richard, Nick).
7. HPD Assembly
Two columns are ready and will be sent for anodizing, prior to transporting to CERN for the HPD dressing. Waiting for quotations for the manufacture of the remaining HPD columns. All components for the cooling plates have been ordered except the honeycomb because the manifold design is not yet complete. The cooling system will be assembled at CERN. The design of the HPD frame is close to final. An overview of the HPD assembly will be presented at the next RICH group meeting next Wednesday (18Oct06). The aim is to be ready for a PRR at the end of Nov06.   
Magnetic shield insulation: Richard P. will meet Tito next week at CERN to set up the lab for wrapping the shields and two Imperial technicians will be at CERN for two weeks to complete as many as possible in the time available.

8.  Magnetic Distortions

Waiting for Ray to report if the system fits.  Ray will be invited to give a report next Wed. at the RICH group meeting.
9.  C4F10
The idea to transport the leftover C4F10 at Oxford to CERN was abandoned because of the very small quantity of C4F10 involved.
10. Aerogel

Need to be discussed further in a dedicated meeting if the aerogel needs to be sealed or not from C4F10 (degradation of clarity).

11. Next meeting
Friday 27th October 2006 at 10:00(GMT) 11:00(CET). 
